Author's response to reviews

Title: Correction of malocclusion and oral rehabilitation in a case of amelogenesis imperfecta by insertion of dental implants followed by Le Fort I distraction osteogenesis of the edentulous atrophic maxilla

Authors:

Aysegul Apaydin (apaydin@istanbul.edu.tr)
Bulent Sermet (bsermet34@yahoo.com)
Sevin Ureturk (Sevinerol@gmail.com)
Abdulsamet Kundakcioglu (abdulsametkundakcioglu@gmail.com)

Version: 10 Date: 9 September 2014

Author's response to reviews: see over
9 September 2014

Dear Editor

I am pleased to resubmit our revised manuscript entitled “Correction of malocclusion and oral rehabilitation in a case of amelogenesis imperfecta by insertion of dental implants followed by Le Fort I distraction osteogenesis of the edentulous atrophic maxilla”, with the alterations which were demanded.

1) The date was changed
2) The patient was diagnosed as amelogenesis imperfect by pediatrics. (Several tests had been made by the pediatrics). She was sent to the pedodontics with that diagnosis for treatment and then to our clinics with the need of surgery. All findings confirmed the initial diagnosis of pediatrics. Additionally, the patient and her family refused to have any extra tests and genetic counseling to confirm the diagnosis once more.

A statement has been added to the text (They are seen in colored)

We hope that our manuscript is appropriate for publication by BMC Oral Health with this form. We hope the language mistakes have been corrected. As you will understand we have to base on the information of a native speaker.

This manuscript has not been published and is not under consideration for publication elsewhere. We have no conflicts of interest to disclose; we do respectfully request that our manuscript is for your consideration for publication.

We look forward to hearing from you at your earliest convenience.

Thank you in advance

Sincerely,

Prof. Dr. Aysegul Apaydin

Professor, Department of Oral and Maxillofacial Surgery